Author’s response to reviews

Title: Delayed Presentation Of A Patient With Sigmoid Colon Injury Following Blunt Abdominal Trauma: A Case Report

Authors:

Gokhan Ertugrul (mdgertugrul@gmail.com)
Murat Coskun (muratcoskuns@yahoo.com.tr)
Mahsuni Sevinc (mahsuni_sevinc@hotmail.com)
Fisun Ertugrul (doctorfisun@mynet.com)
Toygar Toydemir (toygartoydemir@hotmail.com)

Version: 3 Date: 6 January 2012

Author’s response to reviews:

Colonic injury following blunt abdominal trauma is a rare clinical condition. There is no single method to accurately diagnose colon injury caused by blunt abdominal trauma. The rates of morbidity and mortality are high due to delay in diagnosis and treatment.

Herein, we presented a female patient with blunt abdominal trauma following a fall who was admitted to our emergency department in the late period and in whom sigmoid loop colostomy was performed due to sigmoid colon injury; we compared the outcomes with those reported in the literature.